Author's response to reviews

Title: Significance of the parkin and PINK1 gene in Jordanian families with incidences of young-onset and juvenile parkinsonism

Authors:

Ronny Myhre (ronny.myhre@ntnu.no)
Stina Steinkjer (stina.steinkjer@gmail.com)
Alice Stormyr (alsto@hotmail.com)
Gina L Nilsen (gina.l.n@gmail.com)
Hiba A Zayyad (hiba_zayad@yahoo.com)
Khalid Horany (khalidhorany@hotmail.com)
Mohamad K Nusier (mick@just.edu.jo)
Helge Klungland (helge.klungland@ntnu.no)

Version: 6 Date: 3 December 2008

Author's response to reviews: see over
From:
Ronny Myhre
Department of Laboratory Medicine, Children's and Women's Health,
Faculty of Medicine,
NTNU, N-7006 Trondheim,
Norway

To:
The BioMed Central Editorial Production Team 01 December 2008

Dear Production Team,

Please find attached our manuscript, Myhre et al., “Significance of the parkin and PINK1 gene in Jordanian families with incidences of young-onset and juvenile parkinsonism”, with formatting changes as requested.

Major Revisions:

1. Table vertical elements are removed.
2. Figure order is altered so they are sited in consecutive order.
3. Superscript asterisk* have been omitted from figure text.

Minor Revisions:

1. Email addresses are included for all the authors on the title page.
2. The abstract keywords are removed.
3. Figure 2 have been cropped closer to minimise white space around the image.

Please contact me for future correspondence.

Sincerely,
Ronny Myhre